EO-1 Weekly Status Week of February 14 – February 20, 2008
Day of Year 045 - 051
Mission Day 2649 - 2655
Earth Observing-One (EO-1) General

There were 103 Data Collection Events (DCEs) scheduled this week.  
Dan Mandl/581 and Fritz Policelli/604 initiated collaboration with the International Red Cross/Red Crescent (IRC) to begin developing a prototype system to acquire flood imagery and forecasts via use of a flood sensor web to provide to the IRC for use as an emergency response and early warning system.  For example, the IRC would send alerts to targeted areas in Africa to help facilitate evacuation plans.  This collaboration also includes members of the Committee on Earth Observing Satellites (CEOS) subgroup Working Group on Information Systems and Services (WGISS) being led by Martha Maiden from NASA HQ and with satellite teams such as NASA's EO-1, ESA's Envisat and JAXA's ALOS along with flood prediction model teams such as the one from the Ukrainian Space Research Institute. A second coordination telecon was held on Feb 19, 2008 to further define a compelling scenario.  Some preliminary flood products were distributed among the participants and discussions were held to further refine an initial prototype.  The group is attempting to leverage Open Geospatial Consortium (OGC) standards to facilitate interoperability.
EO-1 INSTRUMENTS
All instruments operated nominally this week.
· New ALI flood lamp calibration sequence.

· New sequence generated as an ATS load.

· ATS load successfully up-linked to the spacecraft February 18, 2008 at 20:50z.

· New sequence commands initiated execution on February 19, 2008 at 14:08z.

· New sequence commands completed execution on February 19, 2008 at 14:11z.

· During this time all engineering and science data has been captured through a scheduled GN contact. 

· FOT will generate necessary sequential print and relevant science products to deliver results to the science group.

· Instrument decontamination cycles

· ALI out-gassing performed from Feb 18 (049) at 10:40z to Feb 19 (050) at 01:20z.

· HSI de-icing performed from Feb 18 (049) at 10:30z to Feb 19 (050) at 01:30z.

· HSI solar calibration performed on Feb 19 (050) at 07:38z.

· ALI internal calibration performed on Feb 20 (051) at 15:04z.

EO-1 Spacecraft Subsystems
Command and Data Handling (C&DH)

· Continued to experience problems playing back engineering data from its Solid-State-Recorders.
Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

A teleconference was held between the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday, February 14.  The following items of interest were discussed:
· A list of five Sensor Web activities to be pursued this coming year, for which scheduled action items need to be developed, was discussed.
· The five activities are as follows:

· Participate in the OWS-5 demonstration
· Identify world-wide flood extent information for the International Red Cross 

· Collaborate with the Central American based SERVIR organization that utilizes satellite geospatial data for disaster management 

· Participate in the OGC Empire Challenge 2008

· Develop a public broadcast “Fire Channel” for Southern California - Summer 2008 
· Continued discussion of need to have the SensorML based workflow modified to accept Level 1G data as input.
· Need for the GMU team to modify their BPEL workflow to accept the OAuth authorization security protection and to be able to execute the Vightel Corp. supplied API.
· Vightel Corp. testing of interface to the Northrop Grumman/WUSTL smoke model.
· Need to contact Ernie Hilsenrath (NASA/HQ), CEOS atmospheric chemistry constellation lead, and Jack Fishman (LaRC), atmospheric floods researcher, to explore how to combine CEOS constellations through sensor webs to accomplish GEO societal benefit areas.
· Discussion of the different fire products (i.e. Burn Area Index, Burn Areas Severity, & Hot Pixels Image) and their different usages and the way to produce them using JPL’s web processing service (WPS).
· Need to have OWS-5 demonstration workflow product completed prior to March 26 that includes a video.

· GSFC team is to produce the video storyboard and script.

· Next international flood scenario team (International Red Cross, Japan Aerospace Exploration Agency, Ukrainian flood modeling team, and the GSFC EO-1 sensor web team) telecon to be on February 26.

· GMU team promised to examine/analyze the new interface to the JPL WPS.
· Ames Research Center collaborator is to produce a smoke classifier WCS (Web Coverage Service) for the OWS-5 demonstration.

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week.

EO-1 MISSION OPERATIONS CENTER
Real-Time
· Testing TDRSS downlink times for flood products.

· To determine the usability for the science community FOT, will test downlink rates.

· First TDRSS flood downlink test was conducted on Feb 20 (051) at 19:00z.

· Project aims to bring usable data to the science users within one hour of the image request. 

Mission Planning
No change from last report.

Flight Dynamics
· Flight Dynamics systems updates.

· Parallel testing continues on the new hardware to validate the quality of the products.

· Lunar Calibration parallel testing

· Generation of nominal lunar calibration command sequence completed.

· Generation of modified lunar calibration command sequence completed.

· All three products required for quality assurance have been compared and verified.

· Attitude history file.

· Slew summary file.

· Calibration command file.

· Flight dynamics MATLAB procedures operating as expected.

· Verified visual aid performance of MATLAB function “EO-1 ADS”.

Data Analysis and Trending
No change from last report.
System Administration
· Preparations ongoing for the upcoming EO-1 audit scan.

· An agreed upon date between the FOT and the IT group will be used.

· Mission planning system Speedwheel is down.

· Speedwheel experienced system board failure on February 12, 2008.

· Mission operation’s contractor is pursing purchase of a replacement system board.
· A meeting will be held to discuss and implement modifications to the Sensor Monitor Goal (SGM) code.

· Meeting is scheduled for February 22, 2008 at 16:00z in the EO-1 MOC. 

· FOT is currently developing the proposed code changes.

· System administrators and the ASE-JPL group will be involved in testing and implementation.

Operational Enhancements Requests and Discrepancy Reports
· Proposed set of SGM (Sensor Goal Monitor) code changes are being implemented by the FOT.

· The current operational SGM code has the mailbox file open and the mail-server tries to write the incoming messages received to this file.  It is suspected that, as the SGM code processes the open mailbox file, the mail-server may have problems respecting the lock file generated, and fails to handle the incoming message.

·  The code change would generate a copy of the active mailbox file for processing by the SGM code thereby allowing the incoming deliveries to be made successfully to the original inbox file.  With this change, it is expected that the file handles and permissions will be processed properly.

GROUND AND SPACE NETWORK

Station Downtime
· No major problems to report.

Operational Discrepancies
· No major problems to report.

Support Enhancements  
· Continuing to provide contacts to the SDOGS antennas.
Upcoming Events
· Two Lunar calibrations to be performed.

· Nominal Lunar calibration on February 21, 2008 at 18:03-18:39z.
· Modified Lunar calibration on February 21, 2008 at 22:59-23:33z.

Imagery Status
Scenes and Engineering Cals planned for week of Feb 14 – Feb 20, 2008              103
Total scenes and engineering calibrations planned for entire mission – approximately 48,472
Total Scenes:  ALI scenes in the Level 0 archive            34,895 (as of Feb 20, 2008)

                        Hyperion scenes in the Level 0 archive    34,637
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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